December 18, 2014



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 14768-Author_Edits.doc).

Title: Bleeding and Clotting in Hereditary Hemorrhagic Telangiectasia

Authors: Christopher Dittus, Michael Streiff, Jack Ansell

Name of Journal: World Journal of Clinical Cases

ESPS Manuscript NO: 14768

The manuscript has been improved according to the suggestions of reviewers:

1. Format has been updated

2. Revision has been made according to the suggestions of the reviewer:

- (1) The bleeding risk associated with cerebral AVF, AVM, and telangiectasia was clarified in the text.
- (2) The management of the patient's pulmonary AVMs was added to the case history (these were embolized).
- (3) We feel that the topic of CAD is important, but is beyond the scope of this paper, as is a discussion of antiplatelet agents. Our case, and subsequent review, covers the very specific issue of how to manage HHT patients who have VTE.
- (4) The topic of reversal is interesting and complex. While certainly important, we feel that this discussion opens up too many avenues of discussion that would be too vast for inclusion in this minireview.
- 3. References and typesetting were corrected

Thank you again for publishing our manuscript in the World Journal of Clinical Cases.

Sincerely yours

Christopher Dittus, D.O., M.P.H.

Department of Hematology and Oncology

Boston University Medical Center